Author's response to reviews

Title: Thymidylate Synthase, Dihydropyrimidine Dehydrogenase, ERCC1, and Thymidine Phosphorylase Gene Expression in Primary and Metastatic Gastrointestinal Adenocarcinoma Tissue in Patients Treated on a Phase I Trial of Oxaliplatin and Capecitabine

Authors:

Kazumi Uchida (uchida@hosp.net)
Peter V Danenberg (pdanenbe@usc.edu)
Kathleen D Danenberg (kdanenberg@responsegenetics.com)
Jean L Grem (jgrem@unmc.edu)

Version: 5 Date: 7 December 2008

Author's response to reviews:

1) The fourth table has been renamed Table 4. The appropriate tables are now cited in the manuscript text.
2) Table 2 is now cited in the text.
3) The introduction has been renamed "Background".
4) A conclusions section has been added after the discussion.
5) Table 1 has been revised so that it now fits into a standard portrait page.